Author's response to reviews

Title: Isolated adrenocorticotropic hormone deficiency and thyroiditis associated with nivolumab therapy in a patient with advanced lung adenocarcinoma: A case report and literature review

Authors:

Nobumasa Ohara (oharan@med.niigata-u.ac.jp)
Michi Kobayashi (mt_tagi2011@yahoo.co.jp)
Kazumasa Ohashi (tsukkomi@syd.odn.ne.jp)
Ryo Ito (doragonito2006@yahoo.co.jp)
Yohei Ikeda (ypfranky@yahoo.co.jp)
Gen Kawaguchi (string_k425@yahoo.co.jp)
Yuichiro Yoneoka (yone@bri.niigata-u.ac.jp)
Go Hasegawa (gopatho@med.niigata-u.ac.jp)
Toshinori Takada (ttakada@med.niigata-u.ac.jp)

Version: 2 Date: 25 Jan 2019

Author’s response to reviews:

RESPONSES TO THE REVIEWERS’ COMMENTS

Thank you very much for your excellent advice. Our individual responses to each comment raised are listed below.

Reviewer reports:

Thanks for the revisions.

Give detailed neurological examination on admission. Give information about follow-up for at least 6 months.

Please respond to each comment in your cover letter and highlight the changes you made in response to them in the manuscript.
Authors’ response: Thank you for your advice. We agree and have added the following sentences on detailed neurological examination on admission in the sixth paragraph of the Case Presentation section.

‘No paralysis, cerebellar ataxia, pyramidal or extrapyramidal tract symptoms, visual disturbance, hearing loss, dysarthria, or epileptic seizures were found.’

We have also added the following sentences on the patient’s clinical course after discharge in the twelveth paragraph of the Case Presentation section

‘The subsequent clinical course of the patient was mostly uneventful for 6 months after discharge. As her LAC progressed, her ability to perform activities of daily living decreased, and in November 2018, she was transferred to a local hospital to receive terminal care.’